Dear Dr. Sabina Alam,

Thank you very much for providing us with a chance to revise our manuscript entitled "In-hospital cerebrovascular complications following orthotopic liver transplantation: A retrospective study" (Manuscript ID 5770425052138950). We have now revised the manuscript following the reviewers' kind and constructive suggestions and the changes are listed below at a point-by-point style.

In the present version, we have a thorough language revision done by a English editor. We hope that there is less expressive error in the text. Full title page before abstract with all authors affiliations and emails addresses was added. The tables were included in the main manuscript text after the references and the informed consents and ethical approval were also documented.

We look forward to your favorable consideration.

With best regards,

Sincerely yours,

Jinsheng Zeng, MD, PhD
Department of Neurology and Stroke Center,
First Affiliated Hospital, Sun Yat-Sen University,
No. 58 Zhongshan Road 2, Guangzhou, 510080, P.R. China.

Responses to the reviewers:
Reviewer: Bryan Young
Reviewer's report:
Major Compulsory Revisions: none
Minor Essential Revisions: Grammatical correction needed. For example,
changes "elder" to "older", and "craniotomy" to "craniectomy".
Please clarify if mold/fungal = Aspergillus. If not, what is meant?
Minor correction on page 8, insert a space between "toxoplasma" and "and"

Response: We regret for not revising the manuscript satisfied. In the present version, we have had a native English speaker edited the language of the revision. We hope that there is less expressive error in the text.

Reviewer: J Bradley White
Reviewer's report:
I have reviewed the responses to my requests and believe that they were addressed appropriately. I have no further suggestions.
Response: Thank you very much!